Author’s response to reviews

Title: Validity of Mini-Fluid Challenge for Predicting Fluid Responsiveness Following Liver Transplantation

Authors:

Ahmed Mukhtar (Ahmed.Mukhtar@kasralainy.edu.eg)
Maha Awad (mahaawad23@yahoo.com)
Mohamed Elayashy (Mohamedelayashy@gmail.com)
Amr Hussein (amrhussein1981@hotmail.com)
Gihan Obayah (gihan.obayah@gmail.com)
Akram El Adawy (akrameladawy@hotmail.com)
Mai Ahmed (mai_ahmed_alia244@yahoo.com)
Hisham Dahab (meshodocot@gmail.com)
Ahmed Hasanin (ahmedmohamedhasanin@gmail.com)
Amr Elfouly (amrefouly@yahoo.com)
Mostafa Abd (drmostafabd@yahoo.com)
Amr Abdelaal (aaabdelaal@yahoo.com)
Jean Teboul (jean-louis.teboul@aphp.fr)

Version: 3 Date: 01 Apr 2019

Author’s response to reviews:

Point by point response to the reviewer’s comments:

o The authors’ names were changed to AE1 and AE2 P 15 Line 56

o The clinical trial registry content was moved to Ethic approve and consent to participate section P15 Line 17